Author’s response to reviews

Title: Efficacy and tolerability of sofosbuvir and daclatasvir for treatment of hepatitis C genotype 1 & 3 in patients undergoing hemodialysis- A prospective interventional clinical trial

Authors:
Shafiq Ur Rehman Cheema (shafiqcheema@yahoo.com)
Muhammad Salman Rehman (dr.salmanrehman@hotmail.com)
Ghulam Hussain (ghkhawaja2@hotmail.com)
Sidra Shafiq Cheema (sidra1@yahoo.com)
Nooman Gilani (ngilani@hotmail.com)

Version: 2 Date: 13 Jun 2019

Author’s response to reviews:

1. Textual overlap: Now, specific pin-pointed text is revised to best possibility to overcome textual overlap. Specifically checked with certain tools as well.

2. Abstract formatting: Abstract structure is now limited to 4 sub-sections as suggested, assuming registration of trial & Keywords not part of abstract.